Author’s response to reviews

Title: Life threatening intracerebral haemorrhage following Saw scaled viper (Echis carinatus) envenoming-Authenticated case report from Sri Lanka

Authors:

Vijayabala Jeevagan (jeevaganv@yahoo.com)
C Lakmal Fonseka (fonseka_lakmal@gmail.com)
C Ariaranee Gnanathasan (ariaraneegnanathasan@gmail.com)

Version: 5 Date: 2 April 2013

Author's response to reviews: see over
Thank you for reviewing and accepting manuscript titled ‘Life threatening intracerebral haemorrhage following Saw- scaled viper (*Echis carinatus*) envenoming- Authenticated case report from Sri Lanka’. We would like to thank the reviewer for his comments in response to which the manuscript was thoroughly revised. We have addressed the comments and suggestions, and the items point-by-point is provided below.

1. We have clarified the Figure 1 according to the requirement of the journal and to the instructions provided. It corresponds to an example of the snake detailed in the case report, rather than the actual snake.
2. Title page included with all required information
3. Abstract re-structured and the manuscript was formatted for the journal requirements
4. The manuscript has reviewed by one of our English colleague and any errors were corrected

Thanking your valuable feedback and assistance,

With kind regards,

Vijayabala Jeevagan